**Author's response to reviews**

**Title:** Transcriptional profiles of pilocytic astrocytoma are related to the three different locations but not to the radiological features and clinical outcome

**Authors:**

Krzysztof Zakrzewski (krzysztof.zakrzewski_xl@wp.pl)
Michal Jarzab (mjarzab@io.gliwice.pl)
Malgorzata Oczko-Wojciechowska (gosiaoczko@io.gliwice.pl)
Aleksandra Pfeifer (apfeifer@io.gliwice.pl)
Barbara Jarzab (bjarzab@io.gliwice.pl)
Pawel P Liberski (pawel.liberski@umed.lodz.pl)
Magdalena Zakrzewska (magdalena.zakrzewska@umed.lodz.pl)

**Version:** 2  **Date:** 27 March 2015

**Author's response to reviews:**

Here I revised my manuscript having taken the requested comments and made the required changes i.e.:

- Included line and page numbers.
- Change the place of Ethics statement which now is in the Methods section and includes the name of the body which gave approval and its reference number.